Author's response to reviews

Title: Mutations in epidermal growth factor receptor and K-ras in Chinese patients with colorectal cancer

Authors:

  Zuo Yunxia (yunxiazuo@hotmail.com)
  Cao Jun (caojun1115lemon@yahoo.com.cn)
  Zhu Guanshan (Mike.Zhu@astrazeneca.com)
  Lu Yachao (Peter.Lu@astrazeneca.com)
  Zhou Xueke (zhouxueke@yahoo.com.cn)
  Li Jin (fudanlijin@163.com)

Version: 10 Date: 18 February 2010

Author's response to reviews: see over
1. The list of authors in the manuscript has been corrected according to the style of ‘First name Initial Last name’ and matched with that in the submission system in either style or order.

2. ‘The PCR cycling variables’ of ‘K-ras exon 2 and exon 3’ in the manuscript in page 6 should be ‘K-ras exon 1 and exon 2’ and it has been corrected.

3. (Page 8, line 5)
The sentence ‘In exon 2 and 3 of the K-ras gene, 34 mutations were found in 33 of 101 samples (32.7%)’ has been revised as ‘In exon 1 and 2 of the K-ras gene, 34 mutations were found in 33 of 101 samples (32.7%)’.